Author's response to reviews

Title: Interventions aimed at increasing research use in nursing: a systematic review

Authors:

David S Thompson (dst3@ualberta.ca)
Carole A Estabrooks (carole.estabrooks@ualberta.ca)
Shannon Scott-Findlay (shannon.scott-findlay@ualberta.ca)
Katherine Moore (katherine.moore@ualberta.ca)
Lars Wallin (lars.wallin@karolinska.se)

Version: 7 Date: 11 May 2007

Author's response to reviews:

Dear Dr Eccles,

Thank you. As requested I made the following formatting changes:

1. The search strategy table is now Table 6 and referenced as such in the text.
2. I've inserted space between the references and figure legend.
3. I've removed the header.
4. I've inserted in the authors' contributions that all authors have read and approved the final manuscript.

Best wishes,
David Thompson